

San Joaquin Valley Air Pollution Control District

GOVERNING BOARD

Sam Armentrout, Chair Councilmember, City of Madera

Ronn Dominici, Vice Chair Supervisor, Madera County

FROM:

DATE:

TO:

RE:

SJVUAPCD Governing Board

April 25, 2002

David L. Crow, Executive Director/APCO Project Coordinator: Seyed Sadredin

Mike Maggard Councilmember, City of Bakersfield

Judith G. Case Supervisor, Fresno County

Barbara Patrick Supervisor, Kern County

Tony Barba Supervisor, Kings County

Jerry O'Banion Supervisor, Merced County

Tom Applegate Councilmember, City of Newman

Jack A. Sieglock Supervisor, San Joaquin County

Nick W. Blom Supervisor, Stanislaus County

William Sanders Supervisor, Tulare County

David L. Crow Executive Director/ Air Pollution Control Officer

Northern Region Office 4230 Kiernan Avenue, Suite 130 Modesto, CA 95356-9322 (209) 557-6400 FAX (209) 557-6475

Central Region Office 1990 East Gettysburg Avenue Fresno, CA 93726-0244 (559) 230-6000 FAX (559) 230-6061

Southern Region Office 2700 M Street, Suite 275 Bakersfield, CA 93301-2373 (661) 326-6900 FAX (661) 326-6985

www.valleyair.org

ADOPT PROPOSED AMENDMENTS TO RULE 2201 (NEW AND MODIFIED STATIONARY SOURCE REVIEW); AND RULE 2530 (FEDERALLY ENFORCEABLE POTENTIAL TO EMIT).

RECOMMENDATION:

Adopt proposed amendments to Rule 2201 (New and Modified Stationary Source Review) and Rule 2530 (Federally Enforceable Potential to Emit). Authorize the Chair to sign the attached resolution.

BACKGROUND:

The District is proposing amendments to Rules 2201 and 2530 to add required changes resulting from the reclassification of the San Joaquin Valley's Federal ozone attainment status to severe nonattainment.

On December 10, 2001, the San Joaquin Valley Air Basin was reclassified from "serious" to "severe" nonattainment status for the one-hour National Ambient Air Quality Standard (NAAQS) for ozone. The reclassification was the result of the Valley's failure to attain the ozone NAAQS before the 1999 deadline and was automatically triggered under the Federal Clean Air Act. This reclassification requires changes to Rule 2201 to change the definition of a major source and increase the required emission offset ratio for the construction or modification of major sources.

DISCUSSION:

A. Rule 2201 (New and Modified Stationary Source Review Rule):

The District's NSR program is designed to meet the state and Federal NSR requirements for non-attainment areas. The District NSR Rule (Rule 2201) applies to new and modified stationary sources which emit Nitrogen Oxides (NOx), Carbon Monoxide (CO), Volatile Organic Compounds (VOC), Sulfur Oxides (SOx), Particulate Matter less than ten microns in aerodynamic diameter (PM10), and other pollutants subject to District permit requirements pursuant to District Rule 2010 (Permits Required).

The major amendments to Rule 2201, as proposed, would include the following:

- Lowering of VOC and NOx emission thresholds for defining major sources (Section 3.24) and Title 1 Modification Thresholds (Section 3.38) as required by the reclassification to severe nonattainment of the federal ozone standard;
- Increasing the major source distance offset ratio (Section 4.8) for NOx and VOC from 1.2 to 1.3 as required by the reclassification to severe nonattainment for ozone;
- Removal of Section 3.38.1 from the definition of a "Title I Modification" since that clause is not included in the federal definition. This change is proposed to ensure consistency with EPA regulations.

B. Rule 2530 (Federally Enforceable Potential to Emit):

District Rule 2530 (Federally Enforceable Potential to Emit) provides a way of voluntarily restricting the potential to emit of a stationary source so it may be exempt from the requirement to obtain a Title V permit. A large number of Valley sources have agreed to limit their actual emissions and comply with the requirements of Rule 2530 in order to receive this Title V permit exemption.

The proposed amendment to Rule 2530, required by EPA as a result of the change in attainment status, is:

- A change in Section 5.2.2 of the rule for the reporting requirements for facilities with emissions of NOx and VOC. For a source to be exempt from reporting requirements under the proposed rule, actual NOx or VOC emissions must be less than 6.25 tons per year.
- A change in the emission limitations for NOx and VOC in Section 6.1 of the rule. For a source to be exempt from Title V permitting under the proposed rule, actual NOx or VOC emissions would be limited to 12.5 tons per year.

Cost Effectiveness and Socioeconomic Analysis

Pursuant to state law, the District is required to analyze the cost effectiveness of new rules that implement Best Available Retrofit Control Technology (BARCT). The proposed amendments do not add BARCT requirements, and are therefore not subject to the cost effectiveness analysis mandate. Additionally, state law requires the District to analyze the socioeconomic impacts of any proposed rule amendments that significantly affects air quality or strengthens an emission limitation. The proposed amendments will have neither effect, and are therefore not subject to the socioeconomic analysis mandate.

Environmental Impacts

Pursuant to the California Environmental Quality Act, staff investigated the possible environmental impacts of the proposed amendments. Based on lack of evidence to the contrary, District staff have concluded that the proposed amendments will not have any significant adverse effect on the environment. Staff recommend filing a Notice of Exemption under the provisions of Public Resource Code 15061(b)(3).

Rule Development Process

District staff conducted a round of public workshops to present the proposed amendments and to receive comments from interested parties and affected industries. Notices for the public workshops were published in a general circulation newspaper in each of the eight San Joaquin Valley counties, and mailed to affected sources and interested parties. The workshop notices solicited written comments to be submitted by mail, and identified the names and telephone numbers of the District staff who could answer questions and respond to comments. Information obtained throughout this workshop process has been used to make appropriate changes to the draft rule.

FISCAL IMPACT:

Staff expects no fiscal impact to result from the proposed rule.

Attachments:

Resolution for Rules 2201 and 2530 (4 pages) Proposed Rule 2201 (New and Modified Stationary Source Review) (34 pages) Proposed Rule 2530 (Federally Enforceable Potential to Emit) (12 pages) Final Draft Staff Report (5 pages)

	SJVUAPCD Governing E
	RULE and 2530 (FEDERALLY ENFORCEABLE POTENTIAL TO EMIT)
1	BEFORE THE GOVERNING BOARD OF THE SAN JOAQUIN VALLEY UNIFIED
2	AIR POLLUTION CONTROL DISTRICT
3 4 5	IN THE MATTER OF: PROPOSED RULE 2201 (New and Modified Stationary Source Review) Rule and AMENDED RULE 2530 (Federally Enforceable Potential to Emit)
6 7	WHEREAS, the San Joaquin Valley Unified Air Pollution Control District (District) is a
7	duly constituted unified air pollution control district, as provided in California Health
8 9	and Safety Code Sections 41050 to 40161; and
9 10	WHEREAS, said district is authorized by California Health and Safety Code Section
11	40702 to make and enforce all necessary and proper orders, rules, and regulations to
12	accomplish the purpose of Division 26 of the Health and Safety Code; and
13	WHEREAS, pursuant to Section 107 of the Clean Air Act and pursuant to section
14	39608 of the California Health and Safety Code, the San Joaquin Valley Air Basin has
15	been classified as a nonattainment area for the national and state health-based ozone
16	air quality standards; and
17	WHEREAS, effective December 10, 2001, the United States Environmental
18	Protection Agency (EPA) reclassified the San Joaquin Valley Air Basin as a severe
19	ozone nonattainment area; and
20	WHEREAS, the federal Clean Air Act (CAA), section 182(d) requires severe ozone
21	areas to meet the serious area requirements and additional severe area requirements
22	including, but not limited to, a reduction to 25 ton per year for the VOC and NOx major
23	source threshold and an increase to 1.3 : 1 of the New Source Review emissions
24	offset ratio ; and
25	WHEREAS, changes to Rules 2201 (New and Modified Stationary Source Review)
26	Rule and 2530 (Federally Enforceable Potential to Emit) are necessary to incorporate
27	the additional severe area requirements of the CAA section 182(d); and
28	

1

SJVUAPCD Governing Board ADOPT PROPOSED AMENDMENTS TO RULE 2201 (NEW AND MODIFIED STATIONARY SOURCE REVIEW) RULE and 2530 (FEDERALLY ENFORCEABLE POTENTIAL TO EMIT) April 25, 2002

WHEREAS, a public hearing for the proposed Rules 2201 and 2530 was duly
 noticed for April 25, 2002, in accordance with California Health and Safety Code
 Section 40725.

4 || NOW, THEREFORE, BE IT RESOLVED AS FOLLOWS:

The Governing Board hereby adopts Rules 2201 (New and Modified Stationary
 Source Review) Rule and 2530 (Federally Enforceable Potential to Emit), as set forth
 in the attached hereto and incorporated herein by this reference. Said rules become
 effective on April 25, 2002.

9
2. The Governing Board hereby finds, based on the evidence and information
10
presented at the hearing upon which its decision is based, all notices required to be
given by law have been duly given in accordance with Health and Safety Code
Section 40725, and the Board has allowed public testimony in accordance with Health
and Safety Code Section 40726.

14 3. In connection with the proposed amendments of said rules, the Board makes
15 the following findings as required by California Health and Safety Code Section
16 40727:

a. NECESSITY. The Governing Board finds, based on the staff report,
public and industry testimony, on the record for this rulemaking proceeding that a
need exists for the said rules. The proposed said rules are necessary to meet the
requirements of the Clean Air Act.

b. AUTHORITY. The Governing Board finds that it has the legal authority to
adopt said rules under the California Health & Safety Code Sections 40000 and
40001.

c. CLARITY. The Governing Board finds that said rules are written or
displayed so that their meaning can be easily understood by those persons or
industries directly affected by them.

27 28

2

SJVUAPCD Governing Board ADOPT PROPOSED AMENDMENTS TO RULE 2201 (NEW AND MODIFIED STATIONARY SOURCE REVIEW) RULE and 2530 (FEDERALLY ENFORCEABLE POTENTIAL TO EMIT) April 25, 2002 1 d. CONSISTENCY. The Governing Board finds that said rules are in harmony with, and not in conflict with or contradictory to, existing statutes, court 2 3 decisions, or state or federal regulations. 4 e. NONDUPLICATION. The Governing Board finds that said rules do not 5 impose the same requirements as any existing state or federal regulation. 6 f. **REFERENCE.** The Governing Board finds that said rules implement 7 section 182(d) of the Clean Air Act. 8 4. The District reviewed for possible environmental impacts of the said rules and based on the lack of evidence to the contrary, the project will not result in any 9 10 significant adverse effects to the environment, and a Notice of Exemption is to be prepared and properly noticed pursuant to the California Environmental Quality Act 11 12 Guidelines (CEQA). 5. The APCO is directed to file with all appropriate agencies certified copies of 13 14 this resolution and the rules adopted herein and is directed to maintain a record of this rulemaking proceeding in accordance with California Health and Safety Code Section 15 40728. 16 17 18 19 20 21 22 23 24 25 26 27 28

	SJVUAPCD Governing Bod ADOPT PROPOSED AMENDMENTS TO RULE 2201 (NEW AND MODIFIED STATIONARY SOURCE REVIEW) RULE and 2530 (FEDERALLY ENFORCEABLE POTENTIAL TO EMIT) April 25, 2002	
1	THE FOREGOING was passed and adopted by the following vote of the	
2	Governing Board of the San Joaquin Valley Unified Air Pollution Control District this	
3	25th day of April 2002, to wit:	
4	AYES: Maggard, Case, Barba, Applegate, Patrick, Blom and Dominici.	
5	and boinfirter.	
6		
7	NOES: None.	
8		
9	ABSENT: Sieglock, Sanders, O'Banion and Armentrout.	1
10	SAN JOAQUIN VALLEY UNIFIED	
11		
12	By Sa Junation	
13	Sam Armentrout, Chair	
14	ATTEST: Clerk to the Governing Board	
15	β β	
16	By <u>Chican Omithu</u> Sissy Smith	
17		
18		
19		
20		
21		
22		
23		
24		
25		
26		
27		
28		
	4	
	7	

RULE 2201 NEW AND MODIFIED STATIONARY SOURCE REVIEW RULE (Adopted September 19, 1991; Amended March 11, 1992; Amended October 29, 1992; Amended December 17, 1992; Amended October 21, 1993; Amended June 15, 1995; Amended August 20, 1998; Amended June 21, 2001 but not effective until August 20, 2001; Amended April XX, 2002)

1.0 Purpose

The purpose of this rule is to provide for the following:

- 1.1 The review of new and modified Stationary Sources of air pollution and to provide mechanisms including emission trade-offs by which Authorities to Construct such sources may be granted, without interfering with the attainment or maintenance of Ambient Air Quality Standards; and
- 1.2 No net increase in emissions above specified thresholds from new and modified Stationary Sources of all nonattainment pollutants and their precursors.
- 2.0 Applicability

This rule shall apply to all new stationary sources and all modifications to existing stationary sources which are subject to the District permit requirements and after construction emit or may emit one or more affected pollutant. The requirements of this rule in effect on the date the application is determined to be complete by the APCO shall apply to such application except as provided in Section 2.1.

- 2.1 The requirements of this rule in effect on the date that the rule was most recently amended shall apply to all applications that have not been deemed complete before that date.
- 3.0 Definitions
 - 3.1 Actual Emissions: emissions having occurred from a source, based on source test or monitoring data, actual fuel consumption, and process data. If source test or monitoring data is not available, other appropriate, APCO-approved, emission factors may be used.
 - 3.2 Actual Emissions Reduction (AER): the decrease of actual emissions, compared to the Baseline Period, from an emissions unit and selected for use as emission offsets or ERC banking. AER shall meet the following criteria:
 - 3.2.1 Shall be real, enforceable, quantifiable, surplus, and permanent.
 - 3.2.2 To be considered surplus, AER shall be in excess, at the time the application for an Emission Reduction Credit or an Authority to Construct authorizing such reductions is deemed complete, of any emissions reduction which:

- 3.2.2.1 Is required or encumbered by any laws, rules, regulations, agreements, orders, or
- 3.2.2.2 Is attributed to a control measure noticed for workshop, or proposed or contained in a State Implementation plan, or
- 3.2.2.3 Is proposed in the APCO's adopted air quality plan pursuant to the California Clean Air Act.
- 3.2.3 Emissions reductions attributed to a proposed control measure, which are excluded pursuant to Section 3.2.2.2 and 3.2.2.3 may be reeligible as AER if the control measures identified in the District Air Quality Plan or State Implementation Plan (SIP), are determined not to be necessary for attainment or maintenance of Ambient Air Quality Standards and the APCO and USEPA have approved amendments to the plan or SIP to reflect this determination.
- 3.3 Administrative Change : a change to an existing permit that:
 - 3.3.1 Corrects typographical errors; or
 - 3.3.2 Identifies a change in the name, address, or phone number of any person identified in the permit, or provides a similar minor administrative change at the source; or
 - 3.3.3 Changes the components of emissions monitoring equipment or other components, which have no effect on the quantity of emissions from an emissions unit, or
 - 3.3.4 Allows for the change of ownership or operational control of a source where the APCO determines that no other change is necessary.
- 3.4 Affected Pollutants: those pollutants for which an Ambient Air Quality Standard has been established by the Environmental Protection Agency or by the ARB, and the precursors to such pollutants, and those pollutants regulated by the Environmental Protection Agency under the Federal Clean Air Act or by the ARB under the Health and Safety Code including, but not limited to, VOC, NO_x, SO_x, PM10, CO, and those pollutants which the Environmental Protection Agency, after due process, or the ARB or the APCO, after public hearing, determine may have a significant adverse effect on the environment, the public health, or the public welfare.
- 3.5 Air Quality Improvement Deduction: a 10 percent discount factor applied to Actual Emission Reductions (AER) before the AER is eligible for banking.

- 3.6 Ambient Air Quality Standards: include State and National Ambient Air Quality Standards. (In the inclusion of this rule in the State Implementation Plan, all references in this rule to Ambient Air Quality Standards shall be interpreted as National Ambient Air Quality Standards.)
- 3.7 Baseline Emissions (BE): for a given pollutant, shall be equal to the sum of:
 - 3.7.1 The pre-project Potential to Emit for:
 - 3.7.1.1 Any unit located at a non-Major Source,
 - 3.7.1.2 Any Highly-Utilized Emissions Unit, located at a Major Source, provided that if the unit has a Specific Limiting Condition (SLC), all units combined under the SLC have an average combined annual Actual Emissions during the two consecutive years immediately prior to filing of an application for an Authority to Construct were equal to or greater than 80% of the unit's pre-project SLC limit,
 - 3.7.1.3 Any Fully-Offset Emissions Unit, located at a Major Source, provided that if the unit has a SLC, all units under the SLC also qualify as Fully Offset Emissions Units, or
 - 3.7.1.4 Any Clean Emissions Unit, located at a Major Source, provided that if the unit has a SLC, all units under the SLC also qualify as Clean Emissions Units.
 - 3.7.2 The Historic Actual Emissions (HAE) for emissions units not specified in Section 3.7.1.
- 3.8 Baseline Period: a period of time equal to either
 - 3.8.1 the two consecutive years of operation immediately prior to the submission date of the Complete Application; or
 - 3.8.2 at least two consecutive years within the five years immediately prior to the submission date of the Complete Application if determined by the APCO as more representative of normal source operation; or
 - 3.8.3 a shorter period of at least one year if the emissions unit has not been in operation for two years and this represents the full operational history of the emissions unit, including any replacement units; or
 - 3.8.4 zero years if an emissions unit has been in operation for less than one year (only for use when calculating AER).

- 3.9 Best Available Control Technology (BACT): is the most stringent emission limitation or control technique of the following:
 - 3.9.1 Achieved in practice for such category and class of source;
 - 3.9.2 Contained in any State Implementation Plan approved by the Environmental Protection Agency for such category and class of source. A specific limitation or control technique shall not apply if the owner of the proposed emissions unit demonstrates to the satisfaction of the APCO that such a limitation or control technique is not presently achievable; or
 - 3.9.3 Contained in an applicable federal New Source Performance Standard; or
 - 3.9.4 Any other emission limitation or control technique, including process and equipment changes of basic or control equipment, found by the APCO to be cost effective and technologically feasible for such class or category of sources or for a specific source.
- 3.10 Biomass-fired power cogeneration facility: A facility capable of generating both electrical and thermal power and fueled exclusively on biomass fuels, consisting of at least 90% of one or more of the following constituents: alfalfa, barley, beanstraw, corn, oats, wheat, orchard and vineyard pruning, and forest residues. Grape stems, grape pomace, almond and walnut shells, construction wood waste, urban wood waste, and lawn trimmings are not considered biomass fuels.
- 3.11 Cargo Carriers: trains dedicated to a specific Stationary Source and vessel dockside activities as defined in 45 Federal Register 52696 (August 7, 1980) for vessels dedicated to a specific Stationary Source. Motor vehicles, as defined by the Vehicle Code of the State of California, are not considered Cargo Carriers.
- 3.12 Clean Emissions Unit: an emissions unit that meets one of the following criteria:
 - 3.12.1 The unit is equipped with an emissions control technology with a minimum control efficiency of at least 95% (or at least 85% for lean-burn, internal combustion engines); or
 - 3.12.2 The unit is equipped with emission control technology that meets the requirements for achieved-in-practice BACT as accepted by the APCO during the five years immediately prior to the submission of the complete application.
- 3.13 Complete Application: an application for an Emission Reduction Credit or an Authority to Construct for a new or modified emissions unit which has been evaluated and found to include all information necessary to determine compliance with applicable rules and requirements.

- 3.14 Contemporaneous Period: a period of five consecutive years immediately prior to the date of initiating construction on a new or modified emissions unit.
- 3.15 Contemporaneous Increase in Permitted Emissions: an increase in permitted emissions authorized at a Stationary Source during a Contemporaneous Period.
- 3.16 Contiguous or Adjacent Property: a property consisting of two or more parcels of land with a common point or boundary, or separated solely by a public roadway or other public right-of-way.
- 3.17 Daily Emissions Limitation (DEL): one or more permit conditions which restrict a unit's maximum daily emissions, to a level at or below the emissions associated with the maximum design capacity. A daily emissions limitation must be:
 - 3.17.1 Contained in the latest Authority to Construct and contained in or enforceable by the latest Permit to Operate for the emissions unit; and
 - 3.17.2 Enforceable, in a practical manner, on a daily basis.
- 3.18 Emissions Unit: an identifiable operation or piece of process equipment such as a source operation which emits, may emit, or results in the emissions of any affected pollutant directly or as fugitive emissions.
- 3.19 Fugitive Emissions: emissions that could not reasonably pass through a vent, chimney, stack, or other functionally equivalent opening. Emissions that are not vented through a stack but can reasonably be captured and vented through a stack are not considered Fugitive. Fugitive emissions shall be included in all calculations, except as provided in Section 3.24.
- 3.20 Fully Offset Emissions Unit: for a given pollutant, an emissions unit for which
 - 3.20.1 Offsets have been provided for the unit's full potential to emit; or
 - 3.20.2 Offsets have been provided for the entire stationary source's potential to emit in excess of the offset trigger level; or
 - 3.20.3 Offsets have previously been provided for the stationary source's NSR balance as calculated pursuant to the NSR rule in effect at the time of the offset action, and the emissions unit was installed after the County baseline date as indicated below:

Table 3-1, County Baseline Dates		
COUNTY	BASELINE DATE	
San Joaquin County	May 29, 1979	
Stanislaus County	June 19, 1979	
Merced, Madera, or Kings County	May 21, 1979	
Fresno County Oil Fields	September 20, 1983	
Fresno County all other sources	January 1, 1977	
Tulare County	June 26, 1979	
	September 12, 1979	
Kern County	June 22, 1987 for heavy oil production	
Heavy Oil Production	operations with negative cumulative net	
	emissions change as of June 22, 1987	
Kern County all other sources	December 28, 1976	

Table 3-1, County Baseline Dates

- 3.21 Heavy Oil: crude oil having an American Petroleum Institute gravity of 20 degrees or less as determined by test method ASTM 287-82.
- 3.22 Highly Utilized Emissions Unit: for a given pollutant, an emissions unit for which the average annual Actual Emissions during the two consecutive years immediately prior to filing of an application for an Authority to Construct were equal to or greater than 80% of the unit's pre-project Potential to Emit. The unit must have been in operation for at least two years and, during that entire period, the unit must have complied with all applicable emission limits and performance standards.
- 3.23 Historical Actual Emissions (HAE): Actual Emissions occurring during the Baseline Period, after discounting for:
 - 3.23.1 Any emissions reductions required or encumbered by any laws, rules, regulations, agreements, orders, or permits; and
 - 3.23.2 Any emissions reductions attributed to a control measure noticed for workshop, or proposed or contained in a State Implementation Plan, and
 - 3.23.3 Any emissions reductions proposed in the District air quality plan for attaining the annual reductions required by the California Clean Air Act, and
 - 3.23.4 Any Actual Emissions in excess of those required or encumbered by any laws, rules, regulations, orders, or permits. For units covered by a Specific Limiting Condition (SLC), the total overall HAE for all units covered by SLC must be discounted for any emissions in excess of that allowed by the SLC.

- 3.24 Major Source: for each pollutant, a Stationary Source with post-project emissions or a post-project Stationary Source Potential to Emit (SSPE2), equal to or exceeding one or more of the following threshold values.
 - 3.24.1 For determining major source status, fugitives shall only be included for calculating the air pollutant post-project emissions or SSPE2 if the source is included in the list of source categories identified in the major source definition in 40 CFR Part 70.2, or when determining if a stationary source is a major air toxics source as defined in Rule 2520.

Table 5-2, Major Source Emission Thresholds		
POLLUTANT	THRESHOLD (POUNDS PER YEAR)	
VOC	100,000 <u>50,000</u>	
NOx	100,000 <u>50,000</u>	
СО	200,000	
PM10	140,000	
SOx	140,000	

Table 3-2, Major Source Emission Thresholds

- 3.24.2 For the purpose of determining major source status, the SSPE2 shall not include the quantity of emission reduction credits (ERC) which have been banked since September 19, 1991 for Actual Emissions Reductions that have occurred at the source, and which have not been used on-site. This ERC quantity includes all ERC held as certificates and all emission reduction credits that have been sold or transferred.
- 3.25 Modification:
 - 3.25.1 An action including at least one of the following items:
 - 3.25.1.1 Any change in hours of operation, production rate, or method of operation of an existing emissions unit, which would necessitate a change in permit conditions.
 - 3.25.1.2 Any structural change or addition to an existing emissions unit which would necessitate a change in permit conditions. Routine replacement shall not be considered to be a structural change.
 - 3.25.1.3 An increase in emissions from an emissions unit caused by a modification of the Stationary Source when the emissions unit is not subject to a daily emissions limitation.
 - 3.25.1.4 Addition of any new emissions unit which is subject to District permitting requirements.

- 3.25.1.5 A change in a permit term or condition proposed by an applicant to obtain an exemption from an applicable requirement to which the source would otherwise be subject.
- 3.25.2 A reconstructed Stationary Source shall be treated as a new Stationary Source and not as a modification.
- 3.25.3 Unless previously limited by a permit condition, the following shall not be considered a modification:
 - 3.25.3.1 A change in ownership of an existing emissions unit with valid Permit to Operate provided that the APCO determines that all applicable offset provisions required by the Permit to Operate will be met;
 - 3.25.3.2 A change in ownership of an entire existing Stationary Source with a valid Permit to Operate;
 - 3.25.3.3 A change which consists solely of a transfer of location of an emissions unit within a Stationary Source; or
 - 3.25.3.4 Routine replacement of a whole or partial emissions unit where the replacement part is the same as the original emissions unit in all respects except for the serial number.
- 3.26 Potential to Emit: the maximum capacity of an emissions unit to emit a pollutant under its physical and operational design. Any physical or operational limitation on the capacity of the source to emit a pollutant, including pollution control equipment and restrictions in hours of operation or on the type or amount of material combusted, stored, or processed, shall be treated as part of its design only if the limitation or the effect it would have on emissions is incorporated into the applicable permit as an enforceable permit condition.
- 3.27 PM10: particulate matter with an aerodynamic diameter smaller than or equal to a nominal ten microns as measured by an applicable reference test method or methods found in Article 2, Subchapter 6, Title 17, California Code of Regulations (commencing with Section 94100).
- 3.28 Precursor: a directly emitted air contaminant that, when released into the atmosphere, forms or causes to be formed or contributes to the formation of a secondary air contaminant for which an Ambient Air Quality Standard has been adopted, or whose presence in the atmosphere will contribute to the violation of one or more Ambient Air Quality Standards. The following precursor-secondary air contaminant relationships shall be used for the purposes of this rule:

PRECURSOR	SECONDARY AIR CONTAMINANT	
Valatila Organia Compounda	a. Photochemical oxidants (Ozone)	
Volatile Organic Compounds	b. The organic fraction of PM10	
	a. Nitrogen Dioxide	
Nitrogen Oxides	b. The nitrate fraction of PM10	
	c. Photochemical oxidants (Ozone)	
	a. Sulfur dioxide	
Sulfur Oxides	b. Sulfates	
	c. The sulfate fraction of PM10	

Table 3-3, Precursors

- 3.29 Quarter: for a non-Seasonal Source, this is defined as a calendar quarter. For a Seasonal Source, a quarter is defined as the entire operating season.
- 3.30 Reasonable Further Progress: as defined by the federal Clean Air Act, Section 182(c)(2)(b).
- 3.31 Reconstructed Source: any Stationary Source undergoing reconstruction where the fixed capital cost of the new components exceeds 50% of the fixed capital cost of a comparable, entirely new Stationary Source. Fixed capital cost is the capital needed to provide depreciable components. Reconstructed Source cost shall include only the cost of all emission-producing equipment and associated integral activities at the stationary source. A reconstructed Stationary Source shall be considered a new Stationary Source and not as a modification of an existing Stationary Source.
- 3.32 Routine Replacement: routine replacement in whole or in part of any article, machine, equipment, or other contrivance with a valid District Permit To Operate provided that all of the following conditions are met:
 - 3.32.1 There is no increase in permitted emissions from the Stationary Source.
 - 3.32.2 There is no increase in design capacity, unless an old part is no longer available in which case the replacement can result in a design capacity increase of up to 10%. No change to the permitted throughput or emissions is authorized due to a change in design capacity as part of routine replacement. Such changes shall require application for permit modification.
 - 3.32.2.1 Permitted throughputs are throughput limits upon which emission calculations are, or could be, based.
 - 3.32.2.2 If there are no throughput limiting conditions, permitted throughput shall be a throughput rate which affects emissions.

- 3.32.3 The replacement equipment performs the same function as the equipment being replaced.
- 3.32.4 The replacement does not constitute a Reconstructed Source (as defined by this rule) or Reconstruction (as defined by any applicable New Source Performance Standard). Reconstructed Source cost shall include only the cost of all emission-producing equipment and associated integral activities at the stationary source.
- 3.32.5 When the entire emissions unit is replaced as a routine replacement action, the emissions unit shall either have been addressed by a BARCT rule or shall be equipped with a control device capable of at least 85% emission control.
- 3.33 Seasonal Source: any Stationary Source with more than 90% of its annual emissions occurring within a consecutive 120-day period.
- 3.34 Specific Limiting Condition (SLC): permit terms or conditions, which can be enforced in a practical manner, contained in Authorities to Construct and Permits to Operate and established pursuant to New Source Review provisions that restrict the total overall permitted emissions from two or more emissions units.
- 3.35 Stationary Source: any building, structure, facility, or installation which emits or may emit any affected pollutant directly or as a fugitive emission. Building, structure, facility or installation includes all pollutant emitting activities including emissions units which:
 - 3.35.1 Are under the same or common ownership or operation, or which are owned or operated by entities which are under common control; and
 - 3.35.2 Belong to the same industrial grouping either by virtue of falling within the same two-digit standard industrial classification code or by virtue of being part of a common industrial process, manufacturing process, or connected process involving a common raw material; and
 - 3.35.3 Are located on one or more contiguous or adjacent properties; or
 - 3.35.4 Are located on one or more properties wholly within either the Western Kern County Oil Fields or the Central Kern County Oil Fields or Fresno County Oil Fields and are used for the production of light oil, heavy oil or gas. Notwithstanding the provisions of this definition, light oil production, heavy oil production, and gas production shall constitute separate Stationary Sources.
- 3.36 Stationary Source Project: A single permitting action involving the modification, addition or shutdown of two or more emissions units. If any increase in emissions

from new or modified emissions is permitted based on emission reductions from one or more emissions units included in the stationary source project, the following condition must also be met:

- 3.36.1 The modification or shutdown resulting in the necessary emission reductions shall occur not later than the date of initial operation of the new or modified emissions unit. If the new or modified emissions unit is, in whole or in part, a replacement for an existing emissions unit at the same stationary source, the APCO may allow a maximum of 90 days as a start up period for simultaneous operation of the existing emissions unit and the replacement emissions unit.
- 3.37 Temporary Replacement Emissions Unit (TREU): an emissions unit which is at a Stationary Source for less than 180 days in any twelve month period and replaces an existing emissions unit which is shutdown for maintenance or repair.
 - 3.37.1 The Potential to Emit from a TREU must not exceed the Potential to Emit from the existing emissions unit.
 - 3.37.2 If a TREU is used to replace a TREU, the combined time at the Stationary Source for the two TREU shall not exceed a total of 180 days in any twelve-month period.
 - 3.37.3 An emissions unit not removed from the Stationary Source within 180 days is not a TREU.
- 3.38 Title I Modification: a Stationary Source modification that results in one of the following emission levels. For the purpose of determining if a project is a Title I modification, fugitive emissions shall not be included in the calculation of the SSPE2 unless the source is included in the list of source categories identified in the major source definition in 40 CFR Part 70.2.
 - 3.38.1 for an existing non-Major Source, the post-project Stationary Source Potential to Emit (SSPE2) is increased to equal or exceed one or more of the following threshold values:

POLLUTANT	SSPE2 (POUNDS PER YEAR)
VOC	100,000
NOx	100,000
60	200,000
PM10	140,000
<u>SOx</u>	140,000

Table 3-4, Title 1 Modification Thresholds

3.38.1 for an existing Major Source, the Contemporaneous Increase in Permitted Emissions (CIPE), is equal to or greater than one or more of the following threshold values;

POLLUTANT	CIPE (POUNDS PER YEAR)	
VOC	50,000	
NOx	50,000	
СО	100,000	
PM10	30,000	
SOx	30,000	

Table 3-4, Title 1 CIPE Thresholds

- 3.38.2 Contemporaneous Increase in Permitted Emissions (CIPE) shall be calculated, on a pollutant-by-pollutant basis, as follows:
 - 3.38.2.1 Sum of Potential to Emit in pounds per year for all new emissions units permitted during the period contemporaneous to the proposed Stationary Source project. (For new units covered by a Specific Limiting Condition (SLC), the overall Potential to Emit for all units covered by the SLC shall be used.);
 - 3.38.2.2 Plus the sum of Stationary Source Project Increase in Permitted Emissions (SSIPE) for modifications allowed during the period contemporaneous to the proposed Stationary Source project. (For Stationary Source projects permitted before (date of adoption of this rule) August 20, 2001, the increase in permitted emissions as calculated pursuant to the New Source Review Rule in effect at the time of the modification shall be used.);
 - 3.38.2.3 Plus any on-site Actual Emission Reductions (AER) banked during the Contemporaneous Period for the proposed Stationary Source Project;
 - 3.38.2.4 Minus any on-site AER, occurring during the Contemporaneous Period which have not been banked; transferred to another entity; used as offsets at another Stationary Source; or for which a banking certificate is voluntarily surrendered to the APCO.
- 3.38.4 Stationary Source Project Increase in Permitted Emissions (SSIPE) shall be calculated, on a pollutant-by-pollutant basis, as the sum of Net Emissions Change (NEC), calculated for all emissions units contained in the Stationary Source project. A SSIPE calculated to a negative value shall be set to zero.
- 3.38.5 Net Emissions Change (NEC) shall be calculated, on a pollutant-by-pollutant basis, as follows

PROPOSED

3.38.5.1 For emissions units not covered by a Specific Limiting Condition (SLC):

$$NEC = PE - BE$$

where,

- NEC = Net Emissions Change for each emissions unit, pounds per year
- PE = Post-project potential to emit for each emissions unit, pounds per year
- BE = Baseline emissions for each emissions unit, pounds per year
- 3.38.5.2 For emissions units covered by a Specific Limiting Condition (SLC):

 $NEC_{SLC} = PE_{SLC} - BE_{SLC}$

where,

- NEC_{SLC} = Overall Net Emissions Change for all units covered by the SLC, pounds per year
- PE_{SLC} = Overall post-project Potential to Emit for all units covered by the SLC, pounds per year
- BE_{SLC} = Overall Baseline Emissions for all units covered by the SLC, pounds per year.
- 4.0 Source Requirements
 - 4.1 Best Available Control Technology (BACT): BACT requirements shall be triggered on a pollutant-by-pollutant basis and on an emissions unit-by-emissions unit basis. Unless exempted pursuant to Section 4.2, BACT shall be required for the following actions:
 - 4.1.1 Any new emissions unit or relocation from one Stationary Source to another of an existing emissions unit with a Potential to Emit exceeding two pounds in any one day; and
 - 4.1.2 Modifications to an existing emissions unit with a valid Permit to Operate resulting in an Adjusted Increase in Permitted Emissions (AIPE) exceeding 2.0 pounds in any one day.
 - 4.2 BACT Exemptions: BACT shall not be required for the following:
 - 4.2.1 CO emissions from a new or modified emissions unit at a Stationary Source with a post project Stationary Source Potential to Emit (SSPE2) of less than 200,000 pounds CO per year.
 - 4.2.2 Cargo Carriers;

- 4.2.3 For existing facilities, the installation or modification of an emission control technique performed solely for the purpose of compliance with the requirements of District, State or Federal air pollution control laws, regulations, or orders, as approved by the APCO, shall be exempt from Best Available Control Technology for all air pollutants, provided all of the following conditions are met:
 - 4.2.3.1 There shall be no increase in the physical or operational design of the existing facility, except for those changes to the design needed for the installation or modification of the emission control technique itself;
 - 4.2.3.2 There shall be no increase in the permitted rating or permitted operating schedule of the permitted unit;
 - 4.2.3.3 There shall be no increase in emissions from the stationary source that will cause or contribute to any violation of a National Ambient Air Quality Standard, Prevention of Significant Deterioration increment, or Air Quality Related Value in Class I areas; and
 - 4.2.3.4 The project shall not result in an increase in permitted emissions or potential to emit of more than 25 tons per year of NOx, or 25 tons per year of VOC, or 15 tons per year of SOx, or 15 tons per year of PM-10, or 50 tons per year of CO.
- 4.2.4 New emissions unit or modification of an existing emissions unit for voluntary reduction in emissions, for the sole purpose of generating emission reduction credits. This exemption applies only to the pollutant for which emission reduction credits are obtained. BACT may be required for other affected pollutants;
- 4.2.5 Temporary Replacement Emissions Units;
- 4.2.6 Routine Replacement; or
- 4.2.7 Transfer of location of emissions units within the same stationary source.
- 4.3 Adjusted Increase in Permitted Emissions (AIPE) Calculations: Adjusted Increase in Permitted Emissions shall be calculated as

$$AIPE = PE2 - HAPE$$

where,

- AIPE = Adjusted Increase in Permitted Emissions, pounds per day
- PE2 = the emissions units post project Potential to Emit, pounds per day
- HAPE = the emissions unit's Historically Adjusted Potential to Emit, pounds per day

4.4 Historically Adjusted Potential to Emit (HAPE) Calculations: Historically Adjusted Potential to Emit shall be calculated as

where,

- PE1 = The emissions unit's Potential to Emit prior to modification or relocation
- EF2 = The emissions unit's permitted emission factor for the pollutant after modification or relocation. If EF2 is greater than EF1 then EF2/EF1 shall be set to 1.
- EF1 = The emissions unit's permitted emission factor for the pollutant before the modification or relocation
- 4.5 Emission Offset Requirements:
 - 4.5.1 If emission offset requirements are triggered pursuant to Section 4.5.3, emission offsets shall be provided for net emissions increases resulting from a project. Offset quantities shall be calculated pursuant to Section 4.7.
 - 4.5.2 For Stationary Sources with a quarterly Potential to Emit which remain constant throughout the year, the amount shall be calculated in pounds per year. For Stationary Sources with quarterly Potential to Emit that is not constant throughout the year, and for Seasonal Sources the amount shall be calculated in pounds per quarter.
 - 4.5.3 Offset requirements shall be triggered on a pollutant-by-pollutant basis. Unless exempted pursuant to Section 4.6, offsets shall be required if the post-project Stationary Source Potential to Emit (SSPE2) equals or exceeds the following offset threshold levels:

POLLUTANT	SSPE2 (POUNDS /YEAR)	
NOx	20,000	
VOC	20,000	
CO (non-attainment areas)	30,000	
CO (attainment areas)	200,000	
SOx	54,750	
PM10	29,200	

Table 4-1, Emissions Offset Threshold Levels

- 4.6 Emission Offset Exemptions: Emission offsets shall not be required for the following:
 - 4.6.1 Increases in carbon monoxide in attainment areas if the applicant demonstrates to the satisfaction of the APCO, that the Ambient Air Quality Standards are not violated in the areas to be affected, and such emissions

will be consistent with Reasonable Further Progress, and will not cause or contribute to a violation of Ambient Air Quality Standards;

- 4.6.2 Emergency equipment that is used exclusively as emergency standby equipment for electric power generation or any other emergency equipment as approved by the APCO that does not operate more than 200 hours per year for non-emergency purposes and is not used pursuant to voluntary arrangements with a power supplier to curtail power. Equipment exempted by this section shall maintain a written record of hours of operation and shall have permit conditions limiting non-emergency operation;
- 4.6.3 Portable equipment which is registered as such in accordance with the provisions of Rule 2280 (Portable Equipment Registration) or the Statewide Portable Equipment Registration Program (California Code of Regulation Title 13, Article 5, Sections 2450-2465).
- 4.6.4 On-site soil or groundwater decontamination performed by, under the jurisdiction of, or pursuant to the requirements of an authorized health officer, agricultural commissioner, fire protection officer, or other authorized government officers, provided emissions do not exceed 4,000 pounds per year of any affected pollutant from all emissions units associated with decontamination project;
- 4.6.5 Temporary Replacement Emissions Units.
- 4.6.6 A transfer of location of an entire Stationary Source within the District, under the same owner and provided
 - 4.6.6.1 The Potential to Emit of any affected pollutant will not be greater at the new location than at the previous location when all emissions units are operated at the same permitted conditions; and
 - 4.6.6.2 BACT is applied to all units with emissions equal to or greater than two (2) pounds per day; and
 - 4.6.6.3 The transferred Stationary Source is not added to an existing Stationary Source.
- 4.6.7 A transfer of location of an emissions unit from one Stationary Source to another within the District, under the same owner and provided:
 - 4.6.7.1 The Potential to Emit of any affected pollutant will not be greater at the new location than at the previous location when all emissions units are operated at the same permitted conditions, and

- 4.6.7.2 The offsets that would be otherwise required for the unit at the new location have been provided for the emissions unit previously.
- 4.6.8 For existing facilities, the installation or modification of an emission control technique performed solely for the purpose of compliance with the requirements of District, State or Federal air pollution control laws, regulations, or orders, as approved by the APCO, shall be exempt from offset requirements for all air pollutants provided all of the following conditions are met:
 - 4.6.8.1 There shall be no increase in the physical or operational design of the existing facility, except for those changes to the design needed for the installation or modification of the emission control technique itself;
 - 4.6.8.2 There shall be no increase in the permitted rating or permitted operating schedule of the permitted unit;
 - 4.6.8.3 There shall be no increase in emissions from the stationary source that will cause or contribute to any violation of a National Ambient Air Quality Standard, Prevention of Significant Deterioration increment, or Air Quality Related Value in Class I areas; and
 - 4.6.8.4 The project shall not result in an increase in permitted emissions or potential to emit of more than 25 tons per year of NOx, or 25 tons per year of VOC, or 15 tons per year of SOx, or 15 tons per year of PM-10, or 50 tons per year of CO.
- 4.7 Emission Offset Quantity Calculations:
 - 4.7.1 For pollutants with a pre-project Stationary Source Potential to Emit (SSPE1) greater than the emission offset threshold levels, emission offsets shall be provided for
 - 4.7.1.1 All increases in Stationary Source emissions, calculated as the sum of differences between the post-project Potential to Emit and the Baseline Emissions of all new and modified emissions units, plus
 - 4.7.1.2 All increases in Cargo Carrier emissions.
 - 4.7.2 For pollutants with a pre-project Stationary Source Potential to Emit (SSPE1) below the offset threshold levels, emission offsets shall be provided for

- 4.7.2.1 All increases in Stationary Source emissions above the offset trigger levels, calculated as the difference between the SSPE2 and the offset trigger level, plus
- 4.7.2.2 All increases in Cargo Carrier emissions.
- 4.7.3 The quantity of offsets calculated pursuant to Sections 4.7.1 and 4.7.2 shall be multiplied by the appropriate Distance Offset Ratio to determine the final quantity of offsets required.
- 4.7.4 PM10 Emissions: In determining the quantity of required PM10 offsets, the Total Suspended Particulate Matter (TSP) emissions for which full offsets have been previously provided shall not be recalculated as PM10.
- 4.8 Distance Offset Ratio: For offset calculations, the Distance Offset ratio shall be as shown below:

ORIGINAL LOCATION OF EMISSION OFFSETS	OFFSET RATIO
at the same Stationary Source as the new or modified emissions unit	1.0
within 15 miles of the new or modified emissions unit's Stationary Source	1.2 <u>for Non-Major Sources</u> 1.3 for Major Sources
15 miles or more from the new or modified emissions unit's Stationary Source	1.5

Table 4-2, Distance Offset Ratio

- 4.9 Pre-project Stationary Source Potential to Emit (SSPE1) shall be calculated as the sum of the following:
 - 4.9.1 The Potential to Emit from all units with valid Authorities to Construct (ATC) or Permits to Operate (PTO) at the Stationary Source.
 - 4.9.1.1 For a unit with both a valid ATC and a PTO or a unit with multiple valid ATC, use the ATC or PTO with the highest potential emissions.
 - 4.9.1.2 For units subject to an SLC, the Potential to Emit shall be based on the overall Potential to Emit limit for all units covered by the SLC and not the sum of the individual Potential to Emit of each emissions unit.
 - 4.9.2 The quantity of emission reduction credits (ERC) which have been banked since September 19, 1991 for Actual Emissions Reductions that have occurred at the source, and which have not been used on-site. This quantity includes all ERC held as certificates and all emission reduction

credits that have been sold or transferred. Reductions shall be added to the SSPE1 as positive values.

- 4.10 Post-project Stationary Source Potential to Emit (SSPE2) shall be calculated, on a pollutant-by-pollutant basis, as the sum of the following:
 - 4.10.1 The Potential to Emit from all units with valid Authorities to Construct or Permits to Operate at the Stationary Source, except for emissions units proposed to be shutdown as part of a Stationary Source Project.
 - 4.10.1.1 The Potential to Emit of the post-project Authority to Construct will be used for new or modified units, provided that the ATC will include new conditions canceling the existing ATC or PTO for those units, otherwise use the ATC or PTO with the highest potential emissions.
 - 4.10.1.2 For units subject to an SLC, the Potential to Emit shall be based on the overall Potential to Emit limit for all units covered by the SLC and not the sum of the individual Potential to Emit of each emissions unit.
 - 4.10.2 The quantity of emission reduction credits (ERC) which have been banked since September 19, 1991 for Actual Emissions Reductions that have occurred at the source, and which have not been used on-site. This quantity includes all ERC held as certificates and all emission reduction credits that have been sold or transferred. Reductions shall be added to the SSPE2 as positive values.

4.11 Calculations involving PM10 emissions

- 4.11.1 For existing Stationary Sources for which particulate matter emissions have been calculated as Total Suspended Particulate (TSP), the PM10 emissions shall be recalculated from TSP values using PM10 emission factors or speciation data.
- 4.11.2 In absence of PM10 emissions factors or speciation data, assume 50% of the total suspended particulates is PM10.
- 4.11.3 If the applicant has previously provided full offsets for total suspended particulate matter emissions, those total suspended particulate matter emissions need not be recalculated as PM10, for the purpose of determining the quantity of offsets.

4.12 Actual Emissions Reductions (AER) Calculations: Actual Emissions Reductions shall be calculated, on a pollutant-by-pollutant basis, as follows:

$$AER = HAE - PE2$$

where,

HAE = Historic Actual EmissionsPE2 = Post-project Potential to Emit

- 4.12.1 Prior to banking, AER shall be discounted by 10 percent (10%) for Air Quality Improvement Deduction, and shall comply with all applicable provisions of Rule 2301 (Emission Reduction Credit Banking).
- 4.13 Additional Offset Requirements: Offsets obtained subject to this rule shall comply with the following provisions:
 - 4.13.1 Major Source shutdowns or permanent curtailments in production or operating hours of a Major Source may not be used as offsets for emissions from a Major Source or a Title I modification, unless the ERC, or the emissions from which the ERC are derived, has been included in an EPA-approved attainment plan.
 - 4.13.2 Offsets from another district may be used only if the source of the offsets is within 50 miles of the proposed emissions increases and the APCO has reviewed the permit conditions issued by the district in which the proposed offsets are obtained and certifies that such offsets meet the requirements of this rule and CH&SC Section 40709.6.
 - 4.13.3 Interpollutant offsets may be approved by the APCO on a case-by-case basis, provided that the applicant demonstrates to the satisfaction of the APCO, that the emission increases from the new or modified source will not cause or contribute to a violation of an Ambient Air Quality Standard. In such cases, the APCO shall, based on an air quality analysis, impose offset ratios equal to or greater than the requirements, of this rule.
 - 4.13.3.1 In no case shall exempt compounds or the other compounds excluded from the definition of VOC be used as offsets for VOC.
 - 4.13.3.2 Interpollutant offsets between PM10 and PM10 precursors may be allowed.
 - 4.13.3.3 PM10 emissions shall not be allowed to offset NO_x or reactive organic compound emissions in ozone nonattainment areas, nor be allowed to offset SO_2 emissions in sulfate nonattainment areas.

- 4.13.3.4 Interpollutant offsets between NOx and VOC may be allowed.
- 4.13.4 Actual Emissions Reductions (AER) used as offsets must have occurred during the same calendar quarter as the emissions increases being offset except as allowed pursuant to Sections 4.13.6 through 4.13.9.
- 4.13.5 AER used as offsets for a Seasonal Source must have occurred during the same time period as the proposed source will operate except as allowed pursuant to Sections 4.13.6 through 4.13.9.
- 4.13.6 AER used as offsets for a biomass-fired cogeneration facility may have occurred during any quarter.
- 4.13.7 AER for PM that occurred from October through March, inclusive, may be used to offset increases in PM during any period of the year.
- 4.13.8 AER for NOx and VOC that occurred from April through November may be used to offset increases in NOx and VOC during any period of the year.
- 4.13.9 AER for CO that occurred from November through February may be used to offset increases in CO during any period of the year.
- 4.13.10 AER used as offsets for new and modified Major Sources must be obtained from an area
 - 4.13.10.1 that has a nonattainment classification that is equal to or higher than the area in which the new or modified Major Source is located, and
 - 4.13.10.2 where emissions contribute to a violation of a national Ambient Air Quality Standard in the area in which the new or modified Major Source is located.
- 4.13.11 Offsets required as a condition of an Authority to Construct or a Permit to Operate shall commence not later than the date of initial operation of the new or modified emissions unit.
 - 4.13.11.1 If the new or modified emissions unit is, in whole or in part, a replacement for an existing emissions unit at the same stationary source, the APCO may allow a maximum of 90 days as a start up period for simultaneous operation of the existing emissions unit and the replacement emissions unit.
- 4.13.12 Nothing in this rule shall be construed as requiring ERC used as NSR offsets to be discounted at time of use, except for the additional offsets as required by Sections 4.8 and 4.13.3.

- 4.14 Additional Source Requirements:
 - 4.14.1 Alternative siting: For those sources for which an analysis of alternative sites, sizes, and production processes is required under Section 172 of the Federal Clean Air Act, the applicant shall prepare an analysis functionally equivalent to the requirements of Division 13, Section 21000 et. seq. of the Public Resources Code.
 - 4.14.2 Ambient Air Quality Standards: Emissions from a new or modified Stationary Source shall not cause or make worse the violation of an Ambient Air Quality Standard. In making this determination, the APCO shall take into account the increases in minor and secondary source emissions as well as the mitigation of emissions through offsets obtained pursuant to this rule. Modeling used for the purposes of this rule shall be consistent with the requirements contained in the most recent edition of EPA's "Guideline on Air Quality Models" unless the APCO finds such model is inappropriate for use. After making such a finding, the APCO may designate an alternative model only after allowing for public comments and only with the concurrence of the ARB or the EPA.
 - 4.14.2.1 At the discretion of the APCO, a new or modified source which is not subject to the public noticing requirements of Section 5.4 shall be exempted from the requirements of Section 4.14.2.
 - 4.14.3 Compliance by Other Owned, Operated, or Controlled Source: The owner of a proposed new Major Source or Title I modification shall demonstrate to the satisfaction of the APCO that all major Stationary Sources owned or operated by such person (or by any entity controlling, controlled by, or under common control with such person) in California which are subject to emission limitations are in compliance or on a schedule for compliance with all applicable emission limitations and standards.
- 5.0 Administrative Requirements

The administrative requirements of Sections 5.1 through 5.7, inclusive, shall be applied to all applications for a new or modified emissions unit except for applications for power plants over 50 megawatts. For such power plants, the administrative requirements of Section 5.8 shall apply.

5.1. Complete Application: The APCO shall determine whether the application is complete not later than 30 days after receipt of the application, or after such longer time as both the applicant and the APCO may agree.

- 5.1.1 If the APCO determines that the application is not complete, the applicant shall be notified in writing of the decision specifying the information required. Upon receipt of any resubmittal of the application, a new 30-day period to determine completeness shall begin.
- 5.1.2 Completeness of an application or resubmitted application shall be evaluated on the basis of the information requirements set forth in the District Rules and Regulations as they exist on the date on which the application or resubmitted application is received.
- 5.1.3 Upon determination that the application is complete, the APCO shall notify the applicant in writing.
- 5.1.4 The APCO may, during the processing of the application, request an applicant to clarify, amplify, correct, or otherwise supplement the information submitted in the application.
- 5.2 Preliminary Decision: Following acceptance of an application as complete, the APCO shall perform the evaluations required to determine compliance with this rule and make a preliminary written decision as to whether an Authority to Construct should be approved, conditionally approved, or disapproved.
 - 5.2.1 The APCO shall deny any Authority to Construct if the APCO finds that the subject of the application would not comply with the standards set forth in this rule or any other District rule.
 - 5.2.2 The decision shall be supported by a succinct, written analysis.
- 5.3 Final Action: Within 180 days after acceptance of an application as complete, or within 180 days after the lead agency has approved the project under the California Environmental Quality Act, whichever occurs later, the APCO shall take final action on the application after considering all written comments.
- 5.4 Public Notification and Publication Requirements: The APCO shall provide public notification and publication for the following types of applications:
 - 5.4.1 New Major Sources and Title I modifications.
 - 5.4.2 Applications which include a new emissions unit with a Potential to Emit greater than 100 pounds during any one day for any one pollutant;
 - 5.4.3 Modifications that increase the Stationary Source Potential to Emit (SSPE1) from a level below the emissions offset threshold level to a level exceeding the emissions offset threshold level for one or more pollutants;

- 5.4.4 New Stationary Sources with post-project Stationary Source Potential to Emit (SSPE2) exceeding the emissions offset threshold level for one or more pollutants;
- 5.4.5 Any permitting action resulting in a Stationary Source Project Increase in Permitted Emissions (SSIPE) exceeding 20,000 pounds per year for any one pollutant.
- 5.5 Public Notification and Publication Actions: For the types of applications listed in Section 5.4, the APCO shall perform the following actions:
 - 5.5.1 Within ten (10) calendar days following the preliminary decision the APCO shall publish in at least one newspaper of general circulation in the District a notice stating the preliminary decision, noting how pertinent information can be obtained, and inviting written public comment for a 30 day period following the date of publication.
 - 5.5.2 No later than the date of publication, the APCO shall transmit to the applicant its preliminary written decision, the analysis, and a copy of the notice submitted for publication.
 - 5.5.3 No later than the date of publication, the APCO shall transmit to the USEPA and the CARB and to any person who requests such information, its preliminary written decision, the analysis, and a copy of the notice submitted for publication.
 - 5.5.4 No later than the time the notice of the preliminary decision is published, the APCO shall make available for public inspection at the District office the information submitted by the applicant and the analysis.
 - 5.5.5 The APCO shall provide written notice of the final action to the applicant, the USEPA, and the CARB, and shall publish such notice in a newspaper of general circulation, except that for an application not subject to Section 5.4, the APCO shall not be subject to this section. In such a case, the applicant shall receive notification as provided in Rule 2040 (Applications).
 - 5.5.6 No later than the time of notice of final action is published, the APCO shall make available for public inspection at the District office a copy of the notice submitted for publication and all supporting documents.
- 5.6 Authority to Construct (ATC) General Conditions
 - 5.6.1 An ATC shall not be issued unless the new or modified source complies with the provisions of this rule and all other applicable District Rules and Regulations.

- 5.6.2 An ATC shall require that the new or modified source be built according to the specifications and plans contained in the application.
- 5.6.3 An ATC shall include all those conditions which the APCO deems necessary to assure construction and operation in the manner assumed in making the analysis to determine compliance with this rule.
- 5.6.4 An ATC shall include all those conditions relating to the satisfaction of the offset requirements of this rule.
- 5.6.5 An ATC issued for an emissions unit that relies on reduction in emissions from other units included in the Stationary Source Project, must include a condition that requires initiating and completing construction on those units that provide the reduction prior to commencing operation of the unit with increase in emissions.
 - 5.6.5.1 If the new or modified emissions unit is, in whole or in part, a replacement for an existing emissions unit at the same stationary source, the APCO may allow a maximum of 90 days as a start up period for simultaneous operation of the existing emissions unit and the replacement emissions unit.
- 5.7 Permit to Operate (PTO) General Conditions
 - 5.7.1 A PTO shall require that the new source or modification be operated in the manner assumed in making the analysis to determine compliance with this rule and as conditioned in the Authority to Construct.
 - 5.7.2 A PTO shall include daily emissions limitations and other enforceable conditions which reflect applicable emission limits including the offset requirements.
 - 5.7.3 The APCO shall determine if the applicant has complied with all the conditions in the ATC. The APCO may allow conditions which have not been met at the time the PTO is issued to be incorporated into the Permit to Operate, provided that compliance with that condition is demonstrated by a specified date.
 - 5.7.4 Any source which provides offsets shall be subject to enforceable permit conditions containing specific operational and emissions limitations, which ensure that the emissions reductions will be provided in accordance with the provisions of this rule and shall continue for the reasonably expected life of the proposed source. Where the source of offsets is not subject to a permit, a written contract shall be required between the applicant and the owner of such source, which contract, by its terms, shall be enforceable by the APCO.

The permit and contract shall be submitted to the ARB to be forwarded to the Environmental Protection Agency as part of the State Implementation Plan. A violation of the emission limitation provisions of any such contract shall be chargeable to the applicant.

- 5.7.5 Offsets required as a condition of an ATC or a PTO shall commence not later than the date of initial operation of the new or modified source,
 - 5.7.5.1 If a new or modified Stationary Source is, in whole or in part, a replacement for an existing Stationary Source on the same or contiguous property the APCO may allow a maximum of 90 days as a start up period for simultaneous operation of the existing Stationary Source and the new or replacement source.
- 5.8 Power plants over 50 megawatts: The administrative requirements of this section shall be applied to all power plants over 50 megawatts proposed to be constructed in the District and for which a Notice of Intention (NOI) or Application for Certification (AFC) has been accepted by the California Energy Commission. The APCO may apply for reimbursement of all costs incurred, including lost fees, in order to comply with the provisions of this section.
 - 5.8.1 Intent to Participate and Preliminary Report: Within 14 days of receipt of a NOI, the APCO shall notify the ARB and the California Energy Commission of the APCO's intent to participate in the NOI proceeding. If the APCO chooses to participate in the NOI proceeding, the APCO shall prepare and submit a report to the ARB and the California Energy Commission prior to the conclusion of the nonadjudicatory hearings specified in Section 25509.5 of the Public Resources Code. The report shall include at least:
 - 5.8.1.1 A preliminary specific definition of BACT for the proposed facility.
 - 5.8.1.2 A preliminary discussion of whether there is substantial likelihood that the requirements of this rule and all other District rules can be satisfied by the proposed facility.
 - 5.8.1.3 A preliminary list of conditions which the proposed facility must meet in order to comply with this rule or any other applicable District rules. The preliminary determinations contained in the report shall be as specific as possible within the constraints of the information contained in the NOI.
 - 5.8.2 Equivalency of Application for Certification to Application for Authority to Construct: The APCO shall consider an Application for Certification (AFC)

to be equivalent to an application for an Authority to Construct, and subject, as such, to all definitions and requirements of this rule.

- 5.8.3 Upon receipt of an AFC for a power plant, the APCO shall conduct a Determination of Compliance review. This review shall determine whether an AFC is complete, and within 20 calendar days of receipt of the AFC, the APCO shall so inform the California Energy Commission and the applicant in writing.
 - 5.8.3.1 If the APCO determines that the application is not complete, the information required shall be specified, and the AFC shall be returned to the applicant for resubmittal. Upon receipt of any resubmittal of the application, a new 20 day period to determine completeness shall begin.
 - 5.8.3.2 Completeness of an application or resubmitted application shall be evaluated on the basis of the information requirements set forth in District Rules and Regulations as they exist on the date on which the application or resubmitted application is received.
- 5.8.4 The APCO may request from the applicant any information necessary for the completion of the Determination of Compliance review. If the APCO is unable to obtain the information, the APCO may petition the presiding Commissioner of the California Energy Commission for an order directing the applicant to supply such information.
- 5.8.5 Within 180 days of accepting an AFC as complete, the APCO shall make a preliminary written decision as to whether a Determination of Compliance Certification should be approved, conditionally approved, or disapproved. The APCO shall deny any Determination of Compliance Certification if the APCO finds that the subject of the application would not comply with the standards set forth in this rule or any other District rule. The decision shall be supported by a succinct, written analysis.
- 5.8.6 Notification and Publication actions shall be conducted according to the requirements of Section 5.5.
- 5.8.7 Within 240 days after acceptance of an application as complete, the APCO, after considering all written comments, shall take final action on the application, which action shall consist of the following:
 - 5.8.7.1 the APCO, if all requirements of this rule are met, shall issue and submit to the California Energy Commission a Determination of Compliance, or advise the Commission that a Determination of Compliance cannot be issued.

- 5.8.7.2 public inspection of final action documents shall be provided for in accordance with Section 5.5.6
- 5.8.8 Equivalency of Determination of Compliance to Authority to Construct:: A Determination of Compliance shall confer the same rights and privileges as an Authority to Construct provided that the California Energy Commission approves the Application for Certification and the certificate granted by the Commission includes all conditions of the Determination of Compliance.
- 5.8.9 The APCO shall issue a Permit to Operate to any applicant receiving a certificate from the California Energy Commission pursuant to this rule provided that the construction or modification is in compliance with all conditions of the certificate and of the Determination of Compliance, and provided that the Permit to Operate includes the conditions prescribed in Section 5.7.
- 5.9 Enhanced Administrative Requirement

Application for a certificate of conformity with the procedural requirements of 40 CFR Part 70, shall be subject to the following enhanced administrative requirements in addition to any other applicable administrative requirements of Section 5.0:

- 5.9.1 New Sources and Significant Permit Modifications
 - 5.9.1.1 Public Notification: The APCO shall provide a written notice of the proposed permit and, upon request, copies of the APCO analysis to interested parties. Interested parties shall include affected states, ARB and persons who have requested in writing to be notified. The notice shall also be given by publication in a newspaper of general circulation in the District and by any other means if necessary to assure adequate notice to the affected public. The public shall be given 30 days from the date of publication to submit written comments on the APCO's proposed action.
 - 5.9.1.2 The notice shall provide the following information:
 - 5.9.1.2.1 The identification of the source, the name and address of the permit holder, the activities and emissions change involved in the permit action;

- 5.9.1.2.2 The name and address of the APCO, the name and telephone number of District staff to contact for additional information;
- 5.9.1.2.3 The availability, upon request, of a statement that sets forth the legal and factual basis for the proposed permit conditions;
- 5.9.1.2.4 The location where the public may inspect the Complete Application, the APCO's analysis, the proposed permit, and all relevant supporting materials;
- 5.9.1.2.5 A statement that the public may submit written comments regarding the proposed decision within at least 30 days from the date of publication and a brief description of commenting procedures, and
- 5.9.1.2.6 A statement that members of the public may request the APCO or his designee to preside over a public hearing for the purpose of receiving oral public comment, if a hearing has not already been scheduled. The APCO shall provide notice of any public hearing scheduled to address the proposed decision at least 30 days prior to such hearing;
- 5.9.1.3 The APCO shall provide written response to persons or agencies that submitted written comments which are postmarked by the close of the public notice and comment period. All written comments and responses to such comments shall be kept on file at the District office and made available upon request.
- 5.9.1.4 A copy of the Complete Application, the APCO's analysis and the proposed permit shall be made available at District offices for public review and comment during normal business hours. The APCO's analysis shall include a statement that sets forth the legal and factual basis for the proposed permit conditions, including references to the applicable statutory and regulatory provisions.
- 5.9.1.5 The APCO shall provide written notice to the EPA of the proposed decision along with copies of the proposed permit, the APCO's analysis, the public notice submitted for publication, and all necessary supporting information.
- 5.9.1.6 If the EPA does not object pursuant to Section 5.9.1.9, the APCO shall issue the final permit.

- 5.9.1.7 If the EPA does not object in writing to the APCO's preliminary decision during the EPA's 45 day review period, any person may petition the EPA within 60 days after the expiration of the EPA's 45 day review period. Any such petition shall be based only on objections to the permit that were raised with reasonable specificity during the public comment period, unless the petitioner demonstrates to the EPA that it was impracticable to raise such objections arose after such period. Petitions shall be based on the compliance of the permit provisions with applicable requirements.
- 5.9.1.8 Within 180 days after acceptance of an application as complete, or within 180 days after the lead agency has approved the project under the California Environmental Quality Act, whichever occurs later, the APCO shall take final action on the application after considering all written comments.
- 5.9.1.9 The APCO shall not issue a permit if the EPA objects to its issuance in writing within 45 days of receipt of the APCO's notice of preliminary decision on the proposed permit.
 - 5.9.1.9.1 Any EPA objection shall include a statement of the EPA's reasons for objection and a description of the terms and conditions that the permit must include to respond to the objections. The EPA shall provide the permit applicant a copy of the objection.
 - 5.9.1.9.2 If the APCO fails, within 90 days after the date of EPA's objection, or within 180 days from the date the application was deemed complete plus any extension allowed by the state law, whichever is sooner, to revise and submit a proposed permit in response to the objection, the APCO shall not issue a certification on conformity to Title V.
 - 5.9.1.9.3 If the EPA objects to the permit as a result of a public petition, the APCO shall not issue the permit until EPA's objection has been resolved, except that a petition for review does not stay the effectiveness of a permit or its requirements if the permit was issued after the end of the 45-day review period and prior to an EPA objection. If the APCO has issued a permit prior to receipt of an EPA objection, the EPA will modify, terminate, or revoke such permit, and shall do so consistent with procedures in Section 70.7(g)(4) or

(5)(i) and (ii) of the 40 CFR regulations, and the APCO may thereafter reissue only a revised permit that satisfies EPA objection.

- 5.9.1.9.4 EPA objection shall be limited to compliance with applicable requirements and the requirements of 40 CFR part 70.
- 5.9.2 Minor Permit Modifications
 - 5.9.2.1 Within 5 working days after the receipt of a Complete Application for a minor permit modification, the APCO shall provide notification of the proposed permit modification to the EPA, affected states, and interested parties pursuant to Section 5.9.1.1.
 - 5.9.2.2 The APCO shall not issue a final permit modification until after a 45-day period review of the proposed permit modification by EPA or until EPA has notified the APCO that EPA will not object to issuance of the permit modification, whichever is first.
 - 5.9.2.3 Within 90 days after APCO's receipt of an application for a minor permit modification or 15 days after the end of the EPA's 45-day review, whichever is later, the APCO shall do one of the following:
 - 5.9.2.3.1 Issue the permit as proposed;
 - 5.9.2.3.2 Deny the permit modification application;
 - 5.9.2.3.3 Determine that the requested modification does not meet the minor permit modification criteria and should be reviewed pursuant to the administrative requirements for significant permit modifications; or
 - 5.9.2.3.4 Revise the draft permit modification and transmit the new proposed permit modification to EPA and the affected states.

6.0 Certification of Conformity

A new or modified source subject to the requirements of Rule 2520 may choose to apply for a certificate of conformity with the procedural requirements of 40 CFR Part 70. A certification of conformity will allow changes authorized by the Authority to Construct to be incorporated into the Part 70 permit as administrative permit amendments.

- 6.1 The APCO will issue a written certificate of conformity with the procedural requirements of 40 CFR 70.7 and 70.8, and with the compliance requirements of 40 CFR 70.6(8)(c), if the following conditions are met:
 - 6.1.1 The Authority to Construct is issued in conformance with the Enhanced Administrative Requirements of this rule;
 - 6.1.2 The content of the Authority to Construct issued by the APCO complies with the requirements set forth in Section 9.0 of District Rule 2520 (Federally Mandated Operating Permits);
 - 6.1.3 An application for a certificate of conformity with the requirements of 40 CFR Part 70 is submitted with the application for Authority to Construct. The content of application for the certificate of conformity must comply with the requirements of Sections 7.1 of District Rule 2520 (Federally Mandated Operating Permits);
 - 6.1.4 The Authority to Construct contains a statement of conformity with the requirements of Title V and 40 CFR Part 70;
 - 6.1.5 EPA has not objected to the issuance of the Authority to Construct, or EPA's objections have been resolved to the satisfaction of EPA administrator; and
 - 6.1.6 The Part 70 operating permit being issued will contain the federally enforceable requirements contained in the Authority to Construct.
- 6.2 The certificate of conformity with the procedural requirements of 40 CFR Part 70 is valid as long as the Authority to Construct with which it was issued is valid.
- 6.3 Modifications to an Authority to Construct for which a certificate of conformity has been issued are subject to the administrative requirements of Section 11.0 of District Rule 2520 that apply to permit modifications and changes, as well as the requirements of all District Rules that apply to modifications of Authorities to Construct.
- 7.0 Emission Offset Trading and Reporting
 - 7.1 The APCO shall monitor and report emission offset transactions when the ERC are used for emission offsets in Major Source or Title I modification projects.

Transactions which consist solely of the sale or transfer of ERC shall not be included.

- 7.2 The APCO shall annually report such ERC transactions to the USEPA and CARB. The report shall include an analysis which compares the reductions achieved by the ERC calculation methods and the reductions which would occur using ERC which are adjusted at time of use for prevailing federal, state and District Rules and Regulations.
- 8.0 Application Shield for Routine Replacement
 - 8.1 For a Routine Replacement for which an Authority to Construct is required, the permitted source may continue to operate under an application shield, provided that all of the following conditions are met.
 - 8.1.1 An application for the Routine Replacement has been submitted within seven calendar days of completing the routine replacement.
 - 8.1.2 The source operates in compliance with all applicable requirements of the federal, state, and District rules and regulations.
 - 8.2 When the application has been deemed complete by the APCO, the application shield shall be made effective retroactive from the date of application submittal until the application is either approved or denied,
 - 8.2.1 The application shield is not applicable if the District's final action is delayed due to the failure of the applicant to submit timely information requested by the District. The source must also submit additional information for any requirements that become applicable after a complete application is submitted, but before a PTO is issued.
 - 8.3 The application shield does not exempt the operator from any applicable requirements.
 - 8.4 The application shield applies only to an application for a Routine Replacement and does not authorize any increases to the permitted throughput or emissions due to a change in design capacity as part of a Routine Replacement.

PROPOSED

This page intentionally blank.

•

RULE 2530 FEDERALLY ENFORCEABLE POTENTIAL TO EMIT (Adopted June 15, 1995; Amended <u>April XX, 2002</u>)

1.0 Purpose

The purpose of this rule is to restrict the potential to emit of a stationary source so that the source may be exempt from the requirements of Rule 2520 (Federally Mandated Operating Permits).

2.0 Applicability

This rule shall apply to any stationary source which is a major source of regulated air pollutants or of hazardous air pollutants.

3.0 Definitions

All terms shall retain the definitions provided under Rule 2520 unless otherwise defined in this section.

- 3.1 12-month period: a period of twelve consecutive months determined on a rolling basis with a new 12-month period beginning on the first day of each calendar month.
- 3.2 Actual Emissions: the emissions of a regulated air pollutant from a stationary source for every 12-month period. Valid continuous emission monitoring data or source test data shall be preferentially used to determine actual emissions. In the absence of valid continuous emissions monitoring or source test data, the basis for determining actual emissions shall be: throughput of process materials; throughput of materials stored; usage of materials; data provided in manufacturer's product specifications; material volatile organic compound content reports or laboratory analyses; other information required by this rule and applicable District, state, and federal regulations; or information requested in writing by the District. All calculations of actual emissions shall use EPA, CARB, or District approved methods, including emission factors and assumptions.
- 3.3 HAP: an hazardous air pollutant listed in section 112(b) of the federal Clean Air Act.
- 3.3 Major Source of Hazardous Air Pollutants: a stationary source with potential to emit (including fugitive emissions) 10 tons per year or more of a single HAP, or 25 tons per year or more (including fugitive emissions) of a combination of HAPs. Emissions from any oil or gas production well (with its associated equipment) and emissions from any pipeline compressor station shall not be aggregated with

emissions from other similar units to determine whether such units or stations are major sources of HAPs.

- 3.4 Process Statement: an annual report on permitted emission units from an owner or operator of a stationary source certifying under penalty of perjury the following: throughput of process materials; throughput of materials stored; usage of materials; fuel usage; any available continuous emissions monitoring data; hours of operation; and any other information required by this rule or requested in writing by the District.
- 4.0 General Provisions
 - 4.1 Any stationary source that is in compliance with the applicable requirements of this rule, shall not be considered a major source for the purpose of compliance with District Rule 2520.
 - 4.2 Provision for Air Pollution Control Equipment: The owner or operator of a stationary source may take into account the operation of air pollution control equipment on the capacity of the source to emit an air contaminant if the equipment is required by Federal, State, or District rules and regulations or permit terms and conditions. The owner or operator of the stationary source shall maintain and operate such air pollution control equipment in a manner consistent with good air pollution control practice for minimizing emissions. This provision shall not apply after January 1, 1999 unless such operational limitation is federally enforceable or unless the District Board specifically extends this provision and it is submitted to the U.S. EPA. Such extension shall be valid unless, and until, the U.S. EPA disapproves the extension of this provision.
 - 4.3 Within three years of the effective date of District Rule 2520, the District shall maintain and make available to the public upon request, for each stationary source subject to this rule, information identifying the provisions of this rule applicable to the source.
 - 4.4 This rule shall not relieve any stationary source from complying with requirements pertaining to any otherwise applicable preconstruction permit, or to replace a condition or term of any preconstruction permit, or any provision of a preconstruction permitting program. This does not preclude issuance of any preconstruction permit with conditions or terms necessary to ensure compliance with this rule.
 - 4.5 Unless the owner or operator has chosen to operate the stationary source under an alternative operational limit specified in section 6.2, the owner or operator of a stationary source subject to this rule shall obtain any necessary permits prior to

commencing any physical or operational change or activity which will result in actual emissions that exceed the limits specified in section 6.1.

- 4.6 The owner or operator of a stationary source subject to this rule shall obtain any necessary permits prior to commencing any physical or operational change or activity which will result in an exceedance of an applicable operational limit specified in section 6.2.
- 5.0 Exemptions
 - 5.1 The provisions of this rule shall not apply to a stationary source with valid Part 70 permits, or a stationary source that has applied for Part 70 permit in accordance with timelines required in District Rule 2520. Nothing in this rule shall prevent any stationary source which has had a Part 70 permit from qualifying to comply with this rule in the future in lieu of maintaining an application for a Part 70 permit or upon rescission of a Part 70 permit if the owner or operator demonstrates that the stationary source is in compliance with the applicable requirements of this rule.
 - 5.2 The provisions of this rule shall not apply to stationary source whose actual emissions, throughput, or operation, at any time after the effective date of this rule, exceed the limits specified in this rule, provided that the following conditions are met:
 - 5.2.1 The owner or operator has notified the District at least 30 days prior to any exceedence, and will submit an application for a Part 70 permit, or otherwise obtain federally-enforceable permit limits, and
 - 5.2.2 A complete Part 70 permit application is received by the District, or the permit action to otherwise obtain federally-enforceable limits is completed, within 12 months of the date of notification.
 - 5.3 This rule shall not apply to any stationary source which has a valid operating permit with federally enforceable conditions or other federally enforceable limits limiting its potential to emit to below the applicable threshold(s) for a major source.
 - 5.4 The recordkeeping and reporting requirements of this rule shall not apply to the following:
 - 5.4.1 Any stationary source that emits no more than the following quantity of emissions in any 12-month period:
 - 5.4.1.1 5 tons per year of a regulated air pollutant (excluding HAPs),

- 5.4.1.2 2 tons per year of a single HAP,
- 5.4.1.3 5 tons per year of any combination of HAPs, and
- 5.4.1.4 20 percent of any lesser threshold for a single HAP that the EPA may establish by rule.
- 5.4.2 Any stationary sources for which 90 percent of the stationary sources emissions from the permitted emissions units in every 12-month period are associated with an operation for which the throughput is less than or equal to one of the following quantities:
 - 5.4.2.1 1,400 gallons of any combination of solvent containing material but no more than 550 gallons of any one solventcontaining material, provided that the materials do not contain methyl chloroform (1,1,1-trichloroethane), methylene chloride (dichloromethane), tetrachloroethylene (perchloroethylene), or trichloroethylene;
 - 5.4.2.2 750 gallon of any combination of solvent containing materials where the material contain methyl chloroform (1,1,1-trichloroethane), methylene chloride (dichloromethane), tetrachloroethylene (perchloroethylene), or trichloroethylene, but no more than 300 gallons of any one solvent-containing material;
 - 5.4.2.3 500 gallons of solvent-containing (or volatile organic compound containing) material used at a paint spray unit(s);
 - 5.4.2.3 4,400,000 gallons of gasoline dispensed from equipment with Phase I and Phase II Vapor Recovery Systems;
 - 5.4.2.4 470,000 gallons of gasoline dispensed from equipment without Phase I and Phase II Vapor Recovery Systems;
 - 5.4.2.5 1,400 gallons of gasoline combusted;
 - 5.4.2.6 16,600 gallons of diesel fuel combusted;
 - 5.4.2.7 500,000 gallons of distillate oil combusted; or
 - 5.4.2.8 71,400,000 cubic feet of natural gas combusted.

- 5.5 The reporting requirements of this rule shall not apply to stationary sources which emit in every 12-month period less than or equal to the following quantities:
 - 5.5.1 25 tons per year of sulfur dioxide or carbon monoxide;
 - 5.5.2 $\frac{15-6.25}{6.25}$ tons per year for oxides of nitrogen, <u>or</u> volatile organic compounds; or particulate matter with an aerodynamic diameter less than or equal to 10 microns (PM₁₀)
 - 5.5.3 15 tons per year for particulate matter with an aerodynamic diameter less than or equal to 10 microns (PM_{10}) ;
 - 5.5.4 2.5 tons per year of a single HAP;
 - 5.5.5 6.25 tons per year of any combination of HAPs; and
 - 5.5.6 25 percent of any lesser threshold for a single HAP as the U.S. EPA may establish by rule.
- 5.6 Within 30 days of written request by the District or the EPA, the owner or operator of stationary source claiming exemption form the recordkeeping and reporting requirements of this rule shall demonstrate the stationary source's emissions or throughput are not in excess of the applicable quantities set forth in section 5.4.1, 5.4.2, or 5.5 of this rule.
- 6.0 General Requirements

Each stationary source shall meet either the emission limits prescribed in section 6.1 or the alternative operational limits in section 6.2 of this rule.

6.1 Emission Limits:

In every 12-month period, the stationary source shall not emit more than the following quantity of emissions:

- 6.1.1 25 12.5 tons per year of oxides of nitrogen or volatile organic compounds;
- 6.1.2 35 tons per year of particulate matter with an aerodynamic diameter less than or equal to 10 microns (PM_{10}) ;
- 6.1.3 50 tons per year of carbon monoxide or sulfur dioxide;
- 6.1.4 5 tons per year of a single HAP,

- 6.1.5 12.5 tons per year of any combination of HAPs,
- 6.1.6 50 percent of any lesser threshold for a single HAP as the EPA may establish by rule, and
- 6.1.7 50 percent of the major source threshold for any other regulated air pollutant not listed in 6.1.1 through 6.1.6.
- 6.2 Alternative Operational Limits:

Stationary sources for which 90 percent of the stationary sources emissions from the permitted emissions units in every 12-month period are associated with one of the following operations, shall comply with the following operational limits for that operation:

- 6.2.1 Gasoline Dispensing Facility with Phase I and II Vapor Recovery Systems
 - 6.2.1.1 No more than 7,000,000 gallons of gasoline dispensed in any 12-month period.
- 6.2.2 Degreasing or Solvent-Using Units
 - 6.2.2.1 If the solvents do not include methyl chloroform (1,1,1trichloroethane), methylene chloride (dichloromethane), tetrachloroethylene (perchloroethylene), or trichloroethylene, no more than 5,400 gallons of any combination of solventcontaining materials and 2,200 gallons of any one solventcontaining material is used in any 12-month period; or
 - 6.2.2.2 If the solvents include methyl chloroform (1,1,1trichloroethane), methylene chloride (dichloromethane), tetrachloroethylene (perchloroethylene), or trichloroethylene, no more than 2,900 gallons of any combination of solventcontaining materials and 1,200 gallons of any one solventcontaining material is used in any 12-month period.
- 6.2.3 Paint Spraying Units
 - 6.2.3.1 The total usage rate of all VOC-containing materials, including but not limited to coatings, thinners, reducers, and clean-up solution is no more than 2,100 gallons in any 12-month period.

- 6.2.4 Diesel-Fueled Emergency Stand-by Engines of less than 1,000 Brake Horsepower
 - 6.2.4.1 Fuel usage is no more than 66,000 gallons in any 12-month period.
- 7.0 Recordkeeping and Reporting Requirements
 - 7.1 The recordkeeping and reporting requirements of this rule shall not replace any recordkeeping or reporting requirement contained in an operating permit or in a District, State, or Federal rule or regulation.
 - 7.2 Any stationary source previously exempt from the recordkeeping and reporting requirements of this rule in accordance with section 5.4 of this rule, shall comply with applicable recordkeeping and reporting requirements of this rule if the stationary source exceeds the quantities specified in sections 5.4.1, or 5.4.2 of this rule.
 - 7.3 Immediately after the effective date of this rule, the owner or operator of a stationary source subject to this rule shall comply with any applicable recordkeeping requirements in this section.
 - 7.4 Any additional information requested by the APCO shall be submitted to the APCO within 30 days of the date of request.
 - 7.5 Except for stationary sources operating under the alternative operational limits of this rule, the owner or operator shall comply with the following recordkeeping requirements:
 - 7.5.1 The owner or operator of a stationary source subject to this rule shall keep and maintain records for each permitted emission unit or groups of permitted emission units sufficient to determine actual emissions. Such information shall be summarized in a monthly log, maintained on site by the owner or operator for five years, and be made available to District, CARB, or U.S. EPA staff upon request. (In some cases it may be appropriate to keep records on groups of emissions units which are connected in series. Examples are internal combination engines in the oil fields with a common fuel line, or a series of paint spray booths with a common feed.)

7.5.2 Coating/Solvent Emission Unit

The owner or operator of a stationary source subject to this rule that contain a coating/solvent emission unit or uses a coating, solvent, ink or adhesive shall keep and maintain the following records:

- 7.5.2.1 A current list of all coatings, solvents, inks and adhesives in use. This list shall include: information on the manufacturer, brand, product name or code, VOC content in grams per liter or pounds per gallon, HAPs content in grams per liter or pounds per gallon, or manufacturer's product specifications, material VOC content reports or laboratory analyses providing this information;
- 7.5.2.2 A description of any equipment used during and after coating/solvent application, including type, make and model; maximum design process rate or throughput; control device(s) type and description (if any); and a description of the coating/solvent application/drying method(s) employed;
- 7.5.2.3 A monthly log of the consumption of each solvent (including solvents used in clean-up and surface preparation), coating, ink and adhesive used; and
- 7.5.2.4 All purchase orders, invoices, and other documents to support information in the monthly log.

7.5.3 Organic Liquid Storage Unit

The owner or operator of a stationary source subject to this rule that contains a permitted organic liquid storage unit shall keep and maintain the following records:

- 7.5.3.1 A monthly log identifying the liquid stored and monthly throughput; and
- 7.5.3.2 Information on the tank design and specifications including control equipment.
- 7.5.4 Combustion Emission Unit

The owner or operator of a stationary source subject to this rule that contains a combustion emission unit shall keep and maintain the following records:

- 7.5.4.1 Information on equipment type, make and model, maximum design process rate or maximum power input/output, minimum operating temperature (for thermal oxidizers) and capacity, control device(s) type and description (if any) and all source test information; and
- 7.5.4.2 A monthly log of hours of operation, fuel type, fuel usage, fuel heating value (for non-fossil fuels; in terms of BTU/lb or BTU/gal), percent sulfur for fuel oil and coal, and percent nitrogen for coal.
- 7.5.5 Emission Control Unit

The owner or operator of a stationary source subject to this rule that contains an emission control unit shall keep and maintain the following records:

- 7.5.5.1 Information on equipment type and description, make and model, and emission units served by the control unit;
- 7.5.5.2 Information on equipment design including where applicable: pollutant(s) controlled; control effectiveness; maximum design or rated capacity; inlet and outlet temperatures, and concentrations for each pollutant controlled; catalyst data (type, material, life, volume, space velocity, ammonia injection rate and temperature); baghouse data (design, cleaning method, fabric material, flow rate, air/cloth ratio); electrostatic precipitator data (number of fields, cleaning method, and power input); scrubber data (type, design, sorbent type, pressure drop); other design data as appropriate; all source test information; and
- 7.5.5.3 A monthly log of hours of operation including notation of any control equipment breakdowns, upsets, repairs, maintenance and any other deviations from design parameters.
- 7.5.6 General Emission Unit

The owner or operator of a stationary source subject to this rule that contains an emission unit not included in subsections A, B or C above shall keep and maintain the following records:

- 7.5.6.1 Information on the process and equipment including the following: equipment type, description, make and model; maximum design process rate or throughput; control device(s) type and description (if any);
- 7.5.6.2 Any additional information requested in writing by the APCO;
- 7.5.6.3 A monthly log of operating hours, each raw material used and its amount, each product produced and its production rate; and
- 7.5.6.4 Purchase orders, invoices, and other documents to support information in the monthly log.
- 7.5.7 At the time of annual District review of a permit to operate under Rule 2520, each owner or operator of a stationary source subject to this rule shall submit to the District a process statement. The statement shall be signed by the owner or operator and certify that the information provided is accurate and true.
- 7.6 Upon choosing to operate a stationary source subject to this rule under any one alternative operational limit, the owner or operator shall operate the stationary source in compliance with the alternative operational limit as prescribed in section 6.2 and shall comply with the following recordkeeping and reporting requirements:
 - 7.6.1 The owner or operator shall report within 24 hours to the APCO any exceedance of the alternative operational limit.
 - 7.6.2 The owner or operator shall maintain all purchase orders, invoices, and other documents to support information required to be maintained in a monthly log. Records required under this section shall be maintained on site for five years and be made available to District or U.S. EPA staff upon request.
 - 7.6.3 Gasoline Dispensing Facility Equipment with Phase I and II Vapor Recovery Systems

The owner or operator shall operate the gasoline dispensing equipment in compliance with the following requirements:

7.6.3.1 A monthly log of gallons of gasoline dispensed in the preceding month with a monthly calculation of the total

gallons dispensed in the previous 12 months shall be kept on site.

- 7.6.3.2 A copy of the monthly log shall be submitted to the APCO at the time of annual permit renewal. The owner or operator shall certify that the log is accurate and true.
- 7.6.4 Degreasing or Solvent-Using Unit

The owner or operator shall operate the degreasing or solvent-using unit(s) in compliance with the following requirements:

- 7.6.4.1 A monthly log of amount and type of solvent used in the preceding month with a monthly calculation of the total gallons used in the previous 12 months shall be kept on site.
- 7.6.4.2 A copy of the monthly log shall be submitted to the APCO at the time of annual permit renewal. The owner or operator shall certify that the log is accurate and true.
- 7.6.5 Paint Spraying Unit

The owner or operator shall operate the paint spraying unit(s) in compliance with the following requirements:

- 7.6.5.1 A monthly log of the gallons of VOC-containing materials used in the preceding month with a monthly calculation of the total gallons used in the previous 12 months shall be kept on site.
- 7.6.5.2 A copy of the monthly log shall be submitted to the APCO at the time of annual permit renewal. The owner or operator shall certify that the log is accurate and true.
- 7.6.6 Diesel-Fueled Emergency Standby Engine(s) with Output Less Than 1,000 Brake Horsepower

The owner or operator shall operate the emergency standby engine(s) in compliance with the following requirements:

7.6.6.1 A monthly log of hours of operation, gallons of fuel used, and a monthly calculation of the total hours operated and gallons of fuel used in the previous 12 months shall be kept on site.

7.6.6.2 A copy of the monthly log shall be submitted to the APCO at the time of annual permit renewal. The owner or operator shall certify that the log is accurate and true.

8.0 Violations

- 8.1 Failure to comply with any of the applicable provisions of this rule shall constitute a violation of this rule. Each day during which a violation of this rule occurs is a separate offense.
- 8.2 A stationary source subject to this rule shall be subject to applicable federal requirements for a major source, including Rule 2520 when the conditions specified in either subsections 8.2.1 OR 8.2.2 below, occur:
 - 8.2.1 Commencing on the first day following every 12-month period in which the stationary source exceeds a limit specified in section 6.1 above and any applicable alternative operational limit specified in section 6.2, above, or
 - 8.2.2 Commencing on the first day following every 12-month period in which the owner or operator can not demonstrate that the stationary source is in compliance with the limits in section 6.1 above or any applicable alternative operational limit specified in section 6.2 above.
- 8.3 The APCO shall evaluate a stationary source's compliance with the emission limitations in section 6.1 above as part of the District's annual permit review process required by Health & Safety Code section 42301(e). In performing the evaluation, the APCO shall consider any annual process statement submitted pursuant to section 7.0 reporting requirements. In the absence of valid continuous emission monitoring data or source test data, actual emissions shall be calculated using emissions factors approved by the U.S. EPA, CARB, or the APCO.

Final Draft Staff Report Rule 2201 (New and Modified Stationary Source Review Rule); and Rule 2530 (Federally Enforceable Potential to Emit).

Prepared by George Heinen, Technical Projects Coordinator and Rick McVaigh, Permit Services Manager Reviewed by Seyed Sadredin, Director of Permit Services, Scott Nester, Supervising Air Quality Engineer, and Robert Bashian, Senior Air Quality Inspector

April 25, 2002

I. SUMMARY

A. Reasons for Rule Development and Implementation

The District is proposing amendments to Rules 2201and 2530 to add required changes resulting from the reclassification of the San Joaquin Valley's Federal ozone attainment status to severe nonattainment.

EPA recently finalized their decision to reclassify the San Joaquin Valley Air Basin from "serious" to "severe" nonattainment status for the National Ambient Air Quality Standard (NAAQS) for ozone. Official reclassification was effective on December 10, 2001. The reclassification was the result of the Valley's failure to attain the ozone NAAQS by the 1999 deadline and is automatically triggered under the Federal Clean Air Act (CAA). The CAA also requires such areas to meet the serious area requirements and additional severe area requirements of the CAA, Section 182(d). The additional severe area requirements to Rule 2201 and 2530 including a change the definition of a major VOC and NOx source, an increase to the required emission offset ratio and changes to the reporting and exemption levels for actual emissions.

Drafts of the proposed rule amendments were prepared and published. Staff conducted a series of public workshops to discuss these amendments and to receive comments from the public and industry in March 2001. No comments were received on the proposed reclassification changes during the workshop process. Due to delays in the EPA's reclassification process, District action on the rule changes was held in abeyance until recently. To meet the schedule in the EPA's rulemaking notice, staff propose the amendments for adoption by the District Governing Board in the second guarter of 2002.

8

Final Draft Staff Report: Rule 2201 and Rule 2530

March 21, 2002

II. CURRENT AND PROPOSED REGULATIONS

A. Rule 2201 (New and Modified Stationary Source Review Rule):

The District has not attained Federal and state standards for ozone and PM10 and must therefore implement programs which address current and future emissions. Regulation of new emissions is accomplished through an NSR program that addresses both adequate control and mitigation requirements. The District's NSR program is designed to meet the state and Federal NSR requirements for non-attainment areas. The District NSR Rule (Rule 2201) applies to new and modified stationary sources which emit NOx, CO, VOC, SOx, PM10 and other pollutants subject to District permit requirements pursuant to District Rule 2010 (Permits Required).

Key features of Rule 2201 include:

- Best Available Control Technology (BACT): which mandates emission controls to minimize emission increases above de minimus values;
- Emission offsets: which requires emissions above specified offset threshold levels to be mitigated with either concurrent reductions or emission reduction credits (ERC);
- Public notification: a 30-day period prior to issuance of an Authority to Construct (ATC) to garner comments on projects that result in emissions above specified levels;
- Required elements for Authorities to Construct and Permits to Operate;
- Administrative requirements for the processing of NSR applications.

The major amendments to Rule 2201, as proposed, would include the following:

- Lowering of VOC and NOx emission thresholds for defining major sources (Section 3.24) and Title 1 Modification Thresholds (Section 3.38) as required by the reclassification to severe nonattainment for ozone;
- Increasing the Major Source distance offset ratio (Section 4.8) from 1.2 to 1.3 as required by the reclassification to severe nonattainment for ozone; and
- Removal of Section 3.38.1 from the definition of a "Title I Modification" since that clause is not included in the federal definition. This change is proposed to ensure consistency with EPA regulations.

Final Draft Staff Report: Rule 2201 and Rule 2530

March 21, 2002

B. Rule 2530 (Federally Enforceable Potential to Emit):

District Rule 2530 (Federally Enforceable Potential to Emit) provides a way of restricting the potential to emit of a stationary source so it may be exempt from the requirement to obtain a Title V permit. A large number of Valley sources have agreed to limit their actual emissions and comply with the requirements of Rule 2530 in order to receive this Title V permit exemption.

Key features of Rule 2530 include:

- Provisions stating that sources in compliance with Rule 2530 shall not be subject to the requirement to obtain a Title V permit under District Rule 2520.
- Emission limitations for sources that choose to comply with the rule. For criteria pollutants, the limitations are set at one-half of the major source thresholds. Under Rule 2530, sources agreeing to limit their actual emissions to these levels are not required to obtain Title V permits.
- Recordkeeping and reporting requirements.
- Provisions stating that failure to comply with the applicable provisions of the rule constitutes a violation.

The proposed amendment to Rule 2530, required by EPA as a result of the change in attainment status, is:

- A change in Section 5.2.2 of the rule for the reporting requirements for facilities with emissions of NOx and VOC. For a source to be exempt from reporting requirements under the proposed rule, actual NOx or VOC emissions must be less than 10 tons per year.
- A change in the emission limitations for oxides of nitrogen (NOx) and volatile organic compounds (VOC) in Section 6.1 of the rule. For a source to be exempt from Title V permitting under the proposed rule, actual NOx or VOC emissions would be limited to 12.5 tons per year.

III COST EFFECTIVENESS AND SOCIOECONOMIC IMPACT ANALYSIS

Pursuant to state law, the District is required to analyze the cost effectiveness of new rules that implement Best Available Retrofit Control Technology (BARCT). The proposed amendments do not add BARCT requirements, and are therefore not

Final Draft Staff Report: Rule 2201 and Rule 2530

March 21, 2002

subject to the cost effectiveness analysis mandate. Additionally, state law requires the District to analyze the socioeconomic impacts of any proposed rule amendments that significantly affects air quality or strengthens an emission limitation. The proposed amendments will have neither effect, and are therefore not subject to the socioeconomic analysis mandate.

IV RULE CONSISTENCY ANALYSIS

Pursuant to CH&SC Section 40727.2 (g) a rule consistency analysis of these draft rules is not required. The draft rules do not strengthen emission limits or impose more stringent monitoring, reporting, or recordkeeping requirements.

V. ENVIRONMENTAL EFFECTS

Pursuant to the California Environmental Quality Act, staff investigated the possible environmental impacts of the proposed amendments. Based on lack of evidence to the contrary, District staff have concluded that the proposed amendments will not have any significant adverse effect on the environment. Staff recommend filing a Notice of Exemption under the provisions of Public Resource Code 15061(b)(3).

Attachments:

1. Summary of Significant Comments and Responses

Final Draft Staff Report: Rule 2201 and Rule 2530

March 21, 2002

SUMMARY OF SIGNIFICANT COMMENTS MARCH 21, 2002 DRAFT OF PROPOSED RULES 2201 (NEW AND STATIONARY SOURCE REVIEW) RULE AND RULE 2530 (FEDERALLY ENFORCEABLE POTENTIAL TO EMIT)

EPA Comments: No comments were received.

ARB Comments:

- 1. Comment: Rule 2530, Section 5.5.2 indicates the annual actual emission threshold above which a source must make an annual process report to the District. ARB suggests that the threshold should be 6.25 tons per year for severe nonattainment areas as indicated in the CAPCOA/ARB Model Prohibitory Rule to Limit the Potential to Emit, January 1995.
 - **Response:** That section has been changed to incorporate this comment.

Public Comments: No comments were received.